

JAMES Journal of Advances in Modeling Earth Systems*

RESEARCH ARTICLE

10.1029/2021MS002933

Key Points:

- Sensitivity of microphysics model state to stochastic parameter perturbations is highly time dependent
- Parameter perturbations informed by state-dependent covariance reduce model bias and introduce more variability than uncorrelated sampling

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

T. Vukicevic, tomislava.vukicevic@ff.bg.ac.rs

Citation:

Vukicevic, T., Posselt, D. J., & Stankovic, A. (2022). Sensitivity of modeled microphysics to stochastically perturbed parameters. *Journal of Advances in Modeling Earth Systems*, *14*, e2021MS002933. https://doi. org/10.1029/2021MS002933

Received 2 DEC 2021 Accepted 23 JUN 2022

Author Contributions:

Conceptualization: T. Vukicevic Formal analysis: T. Vukicevic Funding acquisition: T. Vukicevic Investigation: T. Vukicevic Methodology: T. Vukicevic Project Administration: T. Vukicevic Resources: T. Vukicevic Software: T. Vukicevic Supervision: T. Vukicevic Validation: T. Vukicevic Writing – original draft: T. Vukicevic Writing – review & editing: T. Vukicevic

© 2022 The Authors. Journal of Advances in Modeling Earth Systems published by Wiley Periodicals LLC on behalf of American Geophysical Union. This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Sensitivity of Modeled Microphysics to Stochastically Perturbed Parameters

T. Vukicevic¹, D. J. Posselt², and A. Stankovic^{1,3}

¹Faculty of Physics, University of Belgrade, Belgrade, Serbia, ²University of California, Los Angeles, CA, USA, ³Department of Meteorology and Bolin Centre for Climate Research, Stockholm University, Stockholm, Sweden

Abstract This study examines the characteristics of several model parameter perturbation methodologies for ensemble simulations of cloud microphysical processes in convection. A simplified 1D model is used to focus the results on cloud microphysics without the complication of feedbacks to the dynamics and environment. Several parameter perturbation methods are tested, including non-stochastic and stochastic with various distributions and parameter covariance. We find that an ensemble comprised of different time-invariant parameters (non-stochastic) exhibits little bias, but small spread. In addition, its behavior does not respect the time evolution of convection through its various phases. Stochastic parameter (SP) methods in which no inter-parameter covariance is applied produce greater spread, but significant bias. The bias is particularly large for lognormal parameter perturbation distributions. The ensemble spread is retained and the bias reduced when time-varying parameter covariance is applied. In this case, the SP scheme is able to adapt to the time and state-dependent covariance structures and produce ensemble characteristics that are consistent with the specific microphysical processes operating at any given time. The results suggest that SP schemes would benefit from inclusion of parameter covariances, and specifically those that vary with the state of the system. It also suggests that a Normal or LogNormal SP scheme with no covariance may significantly impact the ensemble bias. Finally, the results indicate that high temporal and spatial resolution observations may be needed to characterize the variability in parameter values and covariance.

Plain Language Summary All numerical weather and climate prediction models contain approximate representations (parameterizations) of high resolution cloud processes. Model error results from either inaccurate parameter settings, or from the fact that parameters are static while they should vary in time and space. Previous studies have shown that random variation of model parameters can help to improve predictability of weather and climate. This study examines several methods by which model parameters can be varied in time, and finds that the best results are obtained when parameter variations are consistent with the time evolution of the weather system that is being simulated.

1. Introduction

Using physical parameterizations of micro-to turbulent scale processes in weather and earth system (ES) numerical models is unavoidable. Explicit representation of a full range of underlying degrees of freedom is both unfeasible and undesirable because of intrinsic unpredictability at these scales (Craig & Cohen, 2006). Accordingly, parameterizations of subgrid scale processes are and will remain among leading sources of model uncertainty. Understanding and accounting for the impacts of this uncertainty continues to be one of the outstanding challenges for environmental modeling and prediction.

Over the last two decades, the inclusion of stochastic representation of unresolved variability of sub-grid scale processes in weather and ES models has emerged as a viable approach to representing the intrinsic uncertainties of physical parameterizations in the context of improving probabilistic prediction skill using ensembles (Berner et al., 2011, 2015; Bouttier et al., 2012; Buizza et al., 1999; Charron et al., 2010; Christensen et al., 2015; Davini et al., 2017; Hermoso et al., 2021; Jankov et al., 2017; Leutbecher et al., 2017; Palmer et al., 2005, 2009; Ollinaho et al., 2017; Romine et al., 2014; Sanchez et al., 2015; Stanford et al., 2019; Thompson et al., 2021). The representations of uncertainty using this approach are commonly referred to as stochastic parameterizations (Berner et al., 2017). Stochastic parameterizations may be formulated either by including an empirical stochastic representation of the parameterization uncertainty into a numerical model a posteriori (Berner et al., 2017; Palmer, 2001), or by derivation of stochastic governing equations (Debussche et al., 2012). The latter formulation



is more rigorous but not yet mature for weather and ES modeling applications because it requires development of new numerical models (Berner et al., 2017). Empirical stochastic parameterizations are now widely used in existing weather and ES models.

The most commonly used empirical stochastic parameterization, that is directly related to the uncertainty of the existing physical parameterizations in numerical models, is the Stochastically Perturbed Parameterization Tendency (SPPT; Palmer et al., 2009) scheme. The scheme consists of adding random perturbations to net thermodynamic and dynamic tendencies produced by physical parameterizations at resolvable scales. The perturbations are sampled at every time step by multiplying the unperturbed net tendency by a stochastic temporally and spatially correlated noise. Spatio-temporal variability of the noise is governed by a linear Auto Regressive process model (AR1) applied to a Gaussian correlated spatial noise pattern in the horizontal domain. Applications of the SPPT scheme in global models have produced significant positive impacts on ensemble prediction skill by increasing the ensemble spread and reliability and reducing biases in some variables for prediction ranges from days to seasonal climate (Bouttier et al., 2012; Charron et al., 2010; Leutbecher et al., 2017; Ollinaho et al., 2017; Palmer et al., 2009; Sanchez et al., 2015; Weisheimer et al., 2014). The most significant impacts were found in the tropical regions associated with convective and precipitation processes (Leutbecher et al., 2017). Additionally, positive impacts have been demonstrated in climate model simulations for processes such as the Madden-Julian Oscillation (Leutbecher et al., 2017), El Nino Southern Oscillation (Christensen et al., 2017), and tropical precipitation (Davini et al., 2017). Studies with regional high-resolution models using the SPPT scheme have shown improvements also to short-range ensemble prediction, most notably to the ensemble spread (Baker et al., 2014; Berner et al., 2011, 2015; Jankov et al., 2017, 2019; Romine et al., 2014; Wang et al., 2019).

Although including the SPPT scheme as a stochastic representation of parameterized physics uncertainty in deterministic numerical models has shown clear benefits, it has also been noted that the SPPT could lead to numerical instability, and to physical inconsistencies and imbalances in energy and moisture fluxes locally and globally (Davini et al., 2017; Leutbecher et al., 2017; Ollinaho et al., 2017). These shortcomings have been attributed to the assumption of a uniform uncertainty representation for all parameterizations in the model, implying the same error characteristics for different processes, and to the absence of conservation constraints on the tendency perturbations (Leutbecher et al., 2017). To address the need for a more physically-based approach, recent developments have focused on either the implementation of stochastic schemes for specific processes, such as for example, the convective initiation in boundary layer (Kober & Craig, 2016) and deep convection (Bengtsson et al., 2011, 2019), or on holistic formulations that may apply to different parameterizations by using a single formulation for stochastic perturbation of selected parameters within the deterministic parameterizations (Baker et al., 2014; Christensen et al., 2015; Jankov et al., 2017; McCabe et al., 2016; Ollinaho et al., 2017; Thompson et al., 2021). Within the latter framework, hereafter referred to as the Stochastic Parameters (SP) approach, a parameter is considered to be any tunable scalar physical quantity within an existing deterministic parameterization. In the current study we consider the SP approach, focusing on cloud and precipitation microphysics parameterizations.

The mathematical formulation of an SP scheme is similar in simplicity to an SPPT scheme, in that they both make use of a single stochastic process model for any parameterization, but SP applies it to the parameters within each selected parameterization instead of to the tendencies. In this way, the sources of uncertainty unique to each parameterization are represented while preserving the parameterization's built-in physical consistency (Baker et al., 2014; Ollinaho et al., 2017). As a result, the SP schemes produce physically-based and physically-consistent stochastic variability to the tendencies of the resolved states. Evaluations of the SP schemes in global and regional models have shown positive impacts on the ensemble prediction skill of various degrees depending on the prediction time range, the modeling system, and verification variables (Christensen et al., 2015; Hermoso et al., 2021; Jankov et al., 2017; Leutbecher et al., 2017; McCabe et al., 2016; Ollinaho et al., 2017; Stanford et al., 2019; Thompson et al., 2021; Wang et al., 2019). In comparison to the SPPT, the impacts of the SP schemes on ensemble spread were found to be smaller, but adding value when the two approaches were combined (Christensen et al., 2015; Hermoso et al., 2021; Jankov et al., 2019; Ollinaho et al., 2017). These findings point to two non-exclusive conclusions: the unresolved variability has a broader scope than is contained in the parameter uncertainty, and applications of the SP approach may need to capture a larger diversity of processes (i.e., more parameters and parameterizations within a model) with improved process-specific configurations and better understanding of impact mechanisms to achieve full benefits (Jankov et al., 2019; Leutbecher et al., 2017). The



first conclusion underscores the need for better understanding of the sources of unresolved variability, whereas the second encourages further exploration of the known sources.

Recent studies on the application of the SP schemes to microphysics parameterizations using cloud-permitting resolution regional models provided examples of benefits from including the uncertainty of these processes. Stanford et al. (2019) showed that perturbing the ice microphysics parameters involved in mass-size and fallspeed-size relationships using the SPP (Stochastically Perturbed Parameterization; Ollinaho et al., 2017) scheme resulted in significant impacts on the variability of ice cloud optical depth, rain-rate and precipitation in ensemble simulations of deep convection events. Similarly, Hermoso et al. (2021) demonstrated that the Random Parameters (RP) scheme (McCabe et al., 2016) applied to several microphysics parameters in a two-moment bulk microphysics scheme significantly increased ensemble variability of convective storm evolution for the studied case. Thompson et al. (2021) reported important localized impacts on the precipitation properties including changes to maximum hail-size from perturbing only two microphysics parameters using the SPP scheme. These results clearly point to potential for expanding the scope of relevant impacts by including more processes within the SP framework. Another direction for further development that so far has not received much attention concerns configurations of the SP schemes that involve selection of controlling factors, such as temporal and spatial decorrelation scales and statistical distributions for the parameters. The evaluation and calibration of the SP configurations in the prior studies have focused mostly on enhancing the ensemble prediction skill, in particular the ensemble spread (Christensen et al., 2015; Jankov et al., 2019; Ollinaho et al., 2017; Thompson et al., 2021). Although justified for the purpose, this approach does not consider representativeness of the configurations with respect to the process-level uncertainty that the schemes are meant to capture. This could lead to selections of the controlling factors that are inconsistent with the properties of the physical processes represented by the parameterizations (Ollinaho et al., 2017).

This study addresses evaluation of the SP configurations at the process level, focusing on cloud microphysics parameterizations. Our approach is based on the notion that the parameterization uncertainty is a statistical generalization of the parameterization errors with respect to the true processes and as such could be characterized through evaluation against reference data with a high information content about the processes. Candidate approaches include: using Cloud Resolving Models (CRMs) including Large Eddy Simulation configurations, with more detailed representation of cloud microphysics, and comparison versus observations with high information content about process level outputs from the microphysics parameterizations. Both approaches have strengths and weaknesses. The CRMs produce dynamically consistent high information content data but carry their inherent numerical model uncertainties and due to computational burden are also challenging to apply to a wide range of natural scenarios. They have been used for evaluation and calibration of microphysics parameterizations (Khain et al., 2015) and may be in principle adopted to characterize properties of SP-versions of the parameterizations. That methodology would be similar to the coarse-graining method that was applied to evaluate and propose modifications to the SPPT scheme using CRM simulations (Shutts & Pallares, 2014). We are not aware, however, of any published studies that have used CRM data to calibrate SP-microphysics schemes. The comparison versus observations approach has the strength of using data that represent the actual true state with high accuracy (to the limits of observation errors) and for many scenarios. The primary weakness of this approach is that the data are by design limited to the observable quantities, requiring careful selection of types of observations that could provide sufficient information about the process-level state of the microphysics. In this study the evaluation of the SP-microphysics configurations was considered from the prospective of observability by satellite remote sensing.

As a first step we investigated sensitivity of a microphysics parameterization to different configurations of the SP scheme applied to multiple microphysical parameters within a Lagrangian column model that was used in prior studies to characterize parameter uncertainty by means of multivariate nonlinear inversions using simulated remote sensing observations (Posselt & Vukicevic, 2010; van Lier-Walqui et al., 2012, 2014). The column microphysics model is forced with prescribed time-varying profiles of temperature, humidity and vertical velocity. This modeling framework allowed for investigation of the effect of changes in microphysics parameters on the model output in isolation from any feedback to the cloud-scale dynamics. The test case selected in this study of an idealized representation of mid-latitude squall-line convection is the same as was used in prior studies (Posselt & Bishop, 2012; Posselt et al., 2014; Posselt & Vukicevic, 2010; van Lier-Walqui et al., 2012, 2014). Use of this model in prior studies enabled use of multi-parameter distributions from previous Bayesian inversion as the basis



for setting the statistical properties of parameter distributions used in the SP scheme. Additionally, impacts of the non-stochastic and stochastic multi-parameter representation of parameterization uncertainty on the microphysics model solution and the related satellite-based observable quantities could be directly compared.

The current study addresses the following questions.

What is the microphysics state response to stochastically perturbed parameters ?

What is the sensitivity of observable quantities to the SP configurations ?

What is the impact of using correlated parameter perturbations ?

What is the potential of using microphysics sensitive observations to estimate the SP configuration controls ?

The paper is organized as follows. The 1D column model, the formulation of the SP scheme for the study of configurations, and the design of the sensitivity experiments is presented in Section 2. The experiment results are discussed in Section 3. The summary and conclusions are outlined in Section 4.

2. Methodology

2.1. 1D Column Microphysics Model and Case

Our goal is to explore stochastic parameter perturbations in a system that is low-dimensional and simple, enabling straightforward interpretation of results, and also sophisticated enough to represent the complexity of a realistic three dimensional cloud system. Posselt and Vukicevic (2010) described a single column version of the NASA Goddard Cumulus Ensemble (GCE) model (Tao et al., 2014) that simulates the changes a vertical column of the atmosphere would encounter if it were to be embedded in a three dimensional deep convective system. By applying a base state temperature and water vapor profile representative of a deep convective environment, then imposing perturbations to temperature, water vapor, and vertical motion consistent with convective system evolution from initiation through maturity to dissipation, the model can be made to approximate the behavior of a single column within a fully three dimensional environment. The model microphysical parameterization is consistent with those used in modern numerical weather prediction models, and is a single moment bulk scheme (Lang et al., 2007; Lin et al., 1983; Rutledge & Hobbs, 1983, 1984; Tao et al., 2019) with two liquid (cloud droplets and rain) and three ice (suspended ice particles, falling snow/aggregates, and graupel/hail) hydrometeor species. Clouds interact with longwave and shortwave radiation, and are allowed to be advected in the vertical direction only. No sources or sinks of cloud mass due to horizontal advection are considered.

In the setup described in Posselt and Vukicevic (2010), the model emulates the development of a linear squall line type of convection over a three-hour time period, represented by time evolution of the vertical profiles of cloud droplets, rain, pristine ice, graupel and snow as shown in Figure 1 (respectively panels g–k). This is desirable for our experiments, as the evolution proceeds through two distinct phases: convective (0–90 min), during which there is more intense vertical motion and also more intense rainfall, and stratiform (120–180 min), during which there is more moderate vertical motion and lighter rainfall. The 90–120 min time period represents a transition period in between the two phases. In addition, different cloud microphysical processes are active in convective versus stratiform time periods, with convective times controlled more strongly by liquid microphysics and stratiform times controlled more by ice microphysical parameters. This change in behavior allows us to test how the SP scheme is able to adapt to changes in the degree to which various parameters influence the model output.

The specific parameters we perturb (Table 1) control: the snow and graupel particle fall speed, snow and graupel density, rain particle size distribution, and cloud-to-rain autoconversion threshold. As in Posselt and Vukicevic (2010), we use as output quantities of interest the Precipitation Rate (PR), Liquid and Ice Water Paths (respectively LWP and IWP), and Outgoing Longwave and Shortwave (respectively OLW and OSW) radiation, all output from the model as snapshots at regular times. The model outputs shown in Figure 1 were used as the reference control simulation in this study.

2.2. Sampling of Stochastically Perturbed Parameters

The commonly used SP schemes for the global and regional ensemble models are based on the linear AR1 models. They differ however in the selection of the AR1 model class (Grunwald et al., 2000), the quantity to which the



Journal of Advances in Modeling Earth Systems



Figure 1. Control 1D column microphysics model simulation of column integral satellite observations and hydrometeor vertical profiles: (a) AP, (b) PR, (c) LWP, (d) IWP, (e) OLW, (f) OSW, (g) cloud droplets, (h) rain, (i) pristine ice (j) graupel, and (k) snow.

model is applied and the underlying statistical parameter distribution used for sampling the innovations at every time step. For example, the SPP scheme first formulated by Ollinaho et al. (2017) and since applied in a number of other studies makes use of the Gaussian AR1 model applied to spectral coefficients of 2D horizontal Gaussian correlated noise with the innovations based on either the Log-Normal or Normal univariate distributions (Jankov et al., 2017, 2019; Ollinaho et al., 2017; Stanford et al., 2019; Thompson et al., 2021). Another widely used SP scheme referred to as the Random Parameters (RP) uses a non-Gaussian linear AR1 model applied directly to spatially invariant parameters, with the innovation term based on the bounded uniform distribution (Hermoso et al., 2021; McCabe et al., 2016). In this study the generalized formulation of the Gaussian linear AR1 model

Table 1

Microphysics Parameters Selected for Perturbations in the Sensitivity Experiments With the Corresponding Prescribed Mean Values, Range Boundaries and Standard Deviations Used for Sampling the Parameter Variations by Different Configurations of the SP Scheme

| · · · · · | | | | | | |
|--|-----------|-----------------------|-------|-------|--------|--------|
| Parameter description | Symbol | Units | Mean | Min | Max | σ |
| Snow fall speed coefficient | a_s | cm^{1-b_s} | 200.0 | 63.00 | 400.00 | 27.430 |
| Snow fall speed exponent | b_s | None | 0.3 | 0.10 | 0.60 | 0.042 |
| Graupel fall speed coefficient | a_g | cm^{1-b_g} | 400.0 | 85.00 | 800.00 | 39.494 |
| Graupel fall speed exponent | b_g | None | 0.4 | 0.10 | 0.80 | 0.018 |
| Snow particle density | ρ_s | g cm ⁻³ | 0.2 | 0.05 | 0.40 | 0.024 |
| Graupel particle density | $ ho_{g}$ | g cm ⁻³ | 0.4 | 0.05 | 0.80 | 0.045 |
| Threshold cloud mass mixing ratio for autoconversion to rain | q_{c_o} | $g kg^{-1}$ | 1.0 | 0.001 | 2.0 | 0.112 |
| Slope intercept of the rain particle size distribution | N_{0r} | cm ⁻⁴ | 0.5 | 0.1 | 1.00 | 0.024 |

Note. The standard deviations in the last column correspond to 20% of the values obtained using the inversion data from Posselt and Vukicevic (2010) study.

is used (Grunwald et al., 2000) from which these commonly used schemes could be derived as special cases as shown in the Appendix A.

The Gaussian linear AR1 model is expressed

$$\Phi(t + \Delta t) - \underline{\Phi} = \alpha \left[\Phi(t) - \underline{\Phi} \right] + C_{\Phi}^{1/2} \varepsilon(t)$$
⁽¹⁾

Here Φ is a vector stochastic quantity with mean $\underline{\Phi}$, α is the first order regression coefficient determined by the model time step dt and decorrelation time scale Λ , $C_{\Phi}^{1/2} \varepsilon(t)$ is the innovation term where $C_{\Phi}^{1/2}$ is square root covariance matrix, and $\varepsilon(t)$ is a N(0, I) random vector, with I being the identity matrix. This model constitutes the SP scheme whose configurations would depend on the underlying sampling distribution for the innovations, the associated first and second moment statistics, and the decorrelation time scale Λ .

The Log Normal (*LN*) and Normal (*N*) distributions are considered for representing the variability of microphysics parameters within a range of permissible values (Table 1). Using the *LN* distribution the vector stochastic quantity in the model (Equation 1) would consist of elements $\varphi = Ln\left(\frac{p'}{p}\right)$, where p' and p are respectively the perturbed and unperturbed parameter values, the latter being the constant parameter value used in the deterministic microphysics model. Assuming for simplicity that the parameters are uncorrelated would make φ a random $N\left(\mu, \sigma_{\varphi}^2\right)$ quantity. Using the condition that the mean of the perturbed parameter is equal to the unperturbed value renders $\mu = -\frac{\sigma_{\varphi}^2}{2}$ and $\sigma_{\varphi}^2 = Ln\left(\frac{\sigma_p^2}{p^2} + 1\right)$, where σ_p^2 is the assumed (i.e., the prescribed) parameter variance. Using the *N* distribution would give $\varphi = p'$ with the mean $\mu = p$ and the variance σ_p^2 . When the parameters are assumed correlated a non-diagonal $C_{\Phi}^{1/2}$ would need to be specified for either choice of the distribution. Concerning the influence of the decorrelation time scales in this study the relationship $\alpha = e^{-\frac{dt}{h}}$ is used.

For the sensitivity experiments the second moment statistics for the *LN* and *N* parameter distributions were determined based on the data obtained in the prior study, where the uncertainty of microphysics parameters in the 1D column model was estimated by MCMC (Markov Chain Monte Carlo) inversions using simulated satellite-based observable quantities of PR, IWP, LWP, OSW and OLW radiation (Posselt & Vukicevic, 2010). In that study, the observations 30 min apart were derived from the 1D column model solution equal to the one shown in Figure 1 at 30, 60, 90, 120, and 150 min. The MCMC inversions were performed for 10 microphysical parameters, resulting in a 10-dimensional joint empirical statistical distribution.

The data from the prior study (https://doi.org/10.5281/zenodo.5736940, Posselt et al., 2021) consisting of over 1 million samples of valid parameter vectors (the parameter combinations that correspond to the model solution within the assumed error range for the simulated observations) were used to compute variances and square root covariance matrices via the Cholesky decomposition for subsets of the 7 and 8 parameters considered in this study for the sensitivity experiments. Initial sensitivity experiments using either the *LN* or *N* distribution showed



Table 2

Parameter Sampling Configurations Used for the Sensitivity Experiments

| | Stochastic | Innovation distribution | Covariance | Cov time dependence | Number of parameters | Decorrelation time (hours) |
|----------|------------|-----------------------------------|---|------------------------|----------------------|-------------------------------|
| SP-LN | Yes | Log Normal | Variance only based on inversions PV10 | No | 7 and 8 | 1, 3, 6 and 12 |
| SP-N | Yes | Normal | Variance only based on inversions PV10 | No | 7 and 8 | 1, 3, 6 and 12 |
| SP-NCov | Yes | Normal | Full covariance based on inversions PV10 | No | 7 and 8 | 1, 3, 6 and 12 |
| SP-NCovV | Yes | Normal | Full covariance based on storm-phase dependent inversions | Yes | 7 and 8 | 1, 3, 6 and 12 |
| non-SP | No | Empirical based on inversionsPV10 | Samples from empirical distribution | Time invariant samples | 7 and 8 | N/A |

Note. PV10 refers to the origin of the inversion data obtained in the prior study (Posselt & Vukicevic, 2010). Experiments with 7 and 8 perturbed parameters included, respectively, the first 7 and all parameters shown in Table 1.

that the parameter variances based on the inversion data were too high for the application with these distributions in the SP scheme; they resulted in a high probability of perturbed parameter values that were close to the theoretical boundaries of the parameter ranges. For this reason, the variances were reduced to 20% of the original values. The parameters selected for perturbation in this study together with the corresponding units, value ranges, mean and standard derivations are presented in Table 1. The mean values were set to the "true" model solution that was used to simulate the observations in the prior study. The inversion data were used also to randomly select vector parameter samples for ensemble simulations assuming time-invariant parameters. This configuration was used to compare the stochastic versus non-stochastic parameter perturbations. The non-stochastic configuration is equivalent to using the multi-parameter ensemble strategy (Yussouf & Stensrud, 2012).

2.3. Sensitivity Experiments

The sensitivity experiments with the SP scheme involved 100-member ensemble simulations, in which each member was evolved with a different stochastic sequence of parameter perturbations. Additionally, the same size ensemble simulations with randomly sampled constant-in-time parameters described in the prior section were performed, referred to as non-SP configuration. The SP experiments explored impacts of using different statistical distributions (LN and N) for sampling the innovations, uncorrelated and correlated parameter perturbations, and different decorrelation time scales. The comparison of correlated versus uncorrelated perturbations was performed using only the N distribution. All configurations were applied to 7 and 8 parameter sets and decorrelation times of 1, 3, 6 and 12 hr (Table 2).

Two sets of sensitivity experiments were performed using the correlated parameter sampling: one with time-invariant and other with time-varying covariance. The time-invariant covariance was computed using the prior study data as discussed in the previous section. The data for the time-varying covariance were generated in this study motivated by the initial sensitivity experiments, which showed significant variability of the degree of model sensitivity to the SP scheme depending on the storm phase. The results from the sensitivity experiments are discussed in the following section. Here, we present the approach taken to estimate the storm-phase-dependent covariances. New MCMC inversions were performed using the same algorithm as in Posselt and Vukicevic (2010) for three different observing periods instead of the five observation instances (30, 60, 90, 120, and 150 min) within one three-hour model integration as was done in the prior study. The observing periods were set based on the phases of the simulated storm evolution identified as: convective (30, 60, 90 min), transition to stratiform (90, 120 min) and stratiform (120, 150, 180 min). The new inversions used the same observable quantities as in the prior study (PR, LWP, IWP, OSW and OLW), but three separate observing time periods. This resulted in three different empirical joint parameter distributions. Each corresponding covariance matrix was then applied sequentially in the SP scheme configuration with the time-varying covariance (labeled SP-NCovV in Table 2).

Comparison of correlation matrices corresponding to the prior and new inversion data (Figure 2) indicated significant differences in the relationships between the two representations of the covariance parameter variations for certain parameters. For example, the correlation based on the entire storm evolution (the prior study) between





Figure 2. Parameter correlation matrices based on the Bayesian nonlinear inversions using simulated integral satellite observations for: (a) entire storm evolution (the inversions performed by Posselt and Vukicevic (2010) study), and (b) convective, (c) transition and (d) stratiform phases of the storm evolution (the inversions performed in this study).

the coefficients involved in the fall-speed parametrization for the snow and graupel (respectively a_s and a_g) has significantly higher amplitude (+0.47) and is of the opposite sign than the correlations computed from the new data for either phase of the storm (respectively Figures 2a–2d). Positive correlation between snow and graupel fall speed coefficients results from the constraint that $a_s < a_g$ in the MCMC experiments. As a_s increases, a_g must necessarily increase as well. The correlations corresponding to the different storm phases are all of small amplitude and negative (Figures 2b–2d) with the smallest amplitude during the convective phase (-0.06; Figure 2b) and a gradual increase toward the stratiform phase (Figures 2c and 2d). While the parameter value constraint was also enforced for subset time intervals, the snow and graupel parameters had more limited influence relative to the entire simulation. Also for the parameters involved in the rain processes, the threshold mass mixing ratio for auto-conversion to rain (q_{c0}) and the slope intercept of the rain particle size distribution (N_{or}), the prior study correlation shows near independence between these parameters (Figure 2a), whereas the new data indicate



significant negative correlation (-0.44) during the convective phase (Figure 2b), when the warm rain processes were dominant, and near independence for the remaining phases (Figures 2c and 2d). This is consistent with the findings presented in Posselt and Bishop (2012), who found that parameter distributions change with time, and points to the need to account for the time-dependent joint parameter distributions in stochastic parameter perturbation schemes. Additionally the parameter variances using the phase-dependent inversions were found to be about one order of magnitude smaller than the variances based on the entire storm evolution (not shown). This provided additional justification for reducing the latter variances to 20% of the original values for the application in this study, as discussed in Section 2.2.

The configurations of the SP scheme considered for the sensitivity experiments are summarized in Table 2. For all configurations the updates of the perturbed parameter values at every time step during the 1D microphysics model simulations were constrained by the boundary conditions for each parameter and the physical relation $a_g > a_s$. These were imposed by the method of rejection and resampling: invalid samples were rejected and random resampling was applied till the constraints were satisfied. For the configurations using the correlated innovations the resampling involved drawing the vector samples.

3. Results

The impacts of different configurations of the SP scheme were assessed in terms of the ensemble properties in the observation space and for the state quantities of the 1D column model. The output observable quantities included Accumulated Precipitation (AP), PR, LWP, IWP, OSW and OLW. Note that the AP was not used as an observable in MCMC, but we include it here as it is an outcome of interest for applications; convective storm impact is often as much a function of accumulated precipitation as precipitation rate. The model state quantities included vertical profiles of cloud droplets, rain, pristine ice, snow, and graupel. The impacts in the observation space were measured in terms of time dependent root mean squared error (RMSE) and bias relative to the control simulation and ensemble standard deviation (the spread). The analysis of impacts on the model state quantities was focused on the bias. For all quantities the bias diagnostic was defined as the ensemble mean minus the control (Figure 1).

We note that within the idealized ensemble modeling framework in this study the bias relative to the control simulation represents a mean nonlinear response by the microphysics model, including the transformations into the observations space, to the unbiased stochastic scheme configurations as presented in Section 2.2. This bias measure was used to estimate impacts of the parameter variability represented by the different sampling distributions on the ensemble mean errors in the absence of other factors influencing the errors. The results pertaining to the experiments with 7 perturbed parameters and the decorrelation time scale of 3 hr were selected for the presentation, because the equivalent for the 8-parameter configurations and other decorrelation times were similar. The few notable differences are mentioned in the summary.

3.1. Ensemble Performance in Observation Space

The ensemble diagnostics in the observation space exhibited significant time variability for all SP configurations and high sensitivity to the configurations (Figure 3). The time variability was dependent on storm evolution phases and was most pronounced for the PR and radiation observations (Figures 3b, 3e and 3f). For all configurations the bias and spread for the PR exhibited peaks during the early convective phase and then remained high throughout the stratiform phase. The largest PR bias was associated with the SP-LN configuration. For the OLW the bias and spread were small during the early convective phase, large during the transition phase and moderate after (Figure 3e). The highest bias of 8 Wm² was produced by the SP-LN configuration. Most storm-phase dependent variability in the ensemble diagnostics was found for the OSW with prominent peaks in the bias and spread during the convective, mid-transition and end-stratiform phases (Figure 3f). The maximum in negative bias during the convective phase was associated with a negative bias in liquid water path, while the peak in negative bias during the transition phase was associated with a negative bias in the IWP (Figure 3c). As for the OLW the highest bias amplitudes were produced by the SP-LN configurations (reaching a minimum of -41 Wm² during the early convective phase, Figure 3f). For the IWP, LWP and AP the ensemble spread exhibited steady growth throughout the integration period for all SP configurations, unlike the bias which peaked during the transition phase.





Figure 3. Time dependent ensemble root mean squared error (solid), bias (dotted) and spread (dashed) for different simulated observations: (a) AP, (b) PR, (c) IWP, (d) LWP, (e) OLW and (f) OSW, associated with different configurations of the SP scheme. The configurations (Table 2) are represented by different colors as shown in the legend. The displayed results are for the experiments using the decorrelation time of 3 hr and 7 perturbed parameters.

The sensitivity of the ensemble performance to SP configurations was found to be high for all ensemble performance measures and observation quantities. The ensemble bias was overall highest for the SP-LN and SP-N configurations and smallest for the SP-NCovV, while the SP-NCov configuration performance was in between for all observations. The results for the ensemble spread were more dependent on the observation quantity. The SP-NCov and SP-NCovV configurations showed the highest spread values for the AP and IWP, whereas the SP-N and SP-LN configurations were compatible or slightly better for the other observations.

In contrast to the SP configurations the non-SP experiment showed little dependence of the ensemble performance measures on the storm evolution phases (Figure 3, gray curves). As would be expected by the design of this configuration, the ensemble bias was very small and nearly constant throughout the simulation period, with the exception of the last 30 min where it exhibited a slight growth. The RMSE and spread, which were consequently nearly identical, exhibited steady growth throughout the entire period. The spread was significantly smaller than for the SP experiments until about 125 min, after which it reached compatible values and even exceeded the SP-LN and SP-N configurations for the accumulated precipitation. The results point to the fact that the influence of one or more parameters on the model changes with time during the model evolution as various processes related to those parameters are more or less active, leading to a small sensitivity to storm evolution phases when the parameter perturbations are time invariant. The results also suggest that a higher frequency evaluation (e.g., minutes to hourly) of convective scale ensemble prediction is desirable to achieve better understanding of the differences between the microphysics parameter uncertainty representations.

To better understand the relationships between the ensemble properties and the SP configurations the probability density distributions of perturbed parameter values generated in the course of each experiment were examined. The samples consisted of 100×2161 realizations based on all ensemble members (100) and all 5-s time steps (2161) for each experiment. For easier comparison between different experiments the marginal probability densities shown in Figure 4 were estimated using the Gaussian KDE (Kernel Density Estimation) with standard deviation value of 4. The primary difference between these exit distributions was in the densities near one or both boundary values. The configurations with the high densities at the extremes of the range such as the SP-LN, SP-N tended to produce a highly biased model response (Figure 3). The exit distributions associated with the SP-NCov configuration exhibited similar properties to the SP-N for most parameters, except a_g and b_g for which



Journal of Advances in Modeling Earth Systems



Figure 4. Probability densities of perturbed microphysical parameters based on all ensemble members (100) and 5-s time steps (2161) for experiments using different configurations of the SP scheme. The parameter symbols are shown in the title of each panel and are described in Table 1. The SP scheme configurations (Table 2) are represented by different colors as shown in the legend. The displayed densities are for the experiments where 7 (solid curves) and 8 (dashed curves) parameters were perturbed using decorrelation time of 3 hr. The theoretical and sample mean values for each parameter are marked by colored symbols along the horizontal axis as follows: theoretical mean (black star), 7 perturbed parameters (filled circles), 8 perturbed parameters (filled diamonds). The symbols are staggered in the vertical to avoid overlapping.

the SP-NCov distributions were more symmetrical and with lower densities near the boundaries (Figure 4 blue curves).

In contrast the exit marginal distributions associated with the SP-NCovV configuration which led to the least biased model response had low densities near the range boundaries for all parameters (Figure 4 orange curves). The results suggest that sampling strategies without an imposed covariance structure may lead to more frequent sampling near the extremes, and as a consequence may tend toward a stronger nonlinear biased response. The use of process-level covariance estimates may mitigate this behavior and reduce the incidence of bias.

The results also suggest an inverse relationship for the ensemble spread: high densities near the extremes and mutual independence of parameter perturbations (SP-LN and SP-N configurations) may produce less spread than using the process-dependent co-varied parameters for the same choice of the prescribed entry statistics (the mean and variance values). Stated another way, a restricted range of parameter values produced by the imposition of inter-parameter covariance structures in our experiments led to a larger ensemble spread. We note that the most influential parameters are those that are associated with the snow and graupel fall speeds, and that the coefficients and exponents are strongly correlated (Figure 2). We suspect that ignoring these correlations allows the coefficients and exponents to vary in such a way as to lead to compensating effects, reducing the spread of the ensemble.

The exit distributions also point to the differences between the configurations regarding the impact of the physical constraint $a_s < a_g$ on the sampling. For example, it is clearly evident that using this constraint caused pronounced left and right skewness, respectively, for a_s and a_g distributions for the SP-N configuration (respectively panels a and c in Figure 4, red curves) relative to the Normal distribution that would result from sampling by the linear Gaussian stochastic model according to the expression (Equation 1). As a consequence the parameter perturbations for a_g had a notable bias (shown in Figure 4c by a displacement of red diamond and circle symbols from the control/theoretical-mean marked by black star symbol). Similar but less pronounced tendency for skewed sampling of these parameters was exhibited by the SP-NCov configuration and also SP-NCovV to a lesser degree (Figures 4a and 4c, respectively blue and orange curves and the corresponding mean values). For the SP-LN configuration the impact of the physical constraint was reflected in enhanced skewness for a_s and reduced for a_g . The former was associated with a significant negative bias (Figure 4a, displacement of green diamond and circle from the control). Considering that applying known physical constraints to parameter perturbations is desirable



Journal of Advances in Modeling Earth Systems



Figure 5. Difference between the ensemble mean and control simulation for rain vertical profiles as function of time for different configurations of the SP scheme (Table 2): (a) SP-LN, (b) SP-N, (c) SP-NCov, (d) SP-NCovV and (e) non-SP.

the results point to a potential for mitigating their undesirable impacts on the SP sampling (e.g., introducing or enhancing skewness and bias) by using the process-level covariance estimates with built-in constraints.

3.2. Microphysics Response to SP Configurations

In this section the sensitivity of the various cloud hydrometeor species to the SP configurations is analyzed in terms of the differences between the ensemble mean and the control simulation (i.e., the bias). In contrast to the vertically integrated quantities shown in Figure 3, the cloud droplet, rain, pristine ice, snow, and graupel concentrations are more directly related to changes in microphysical parameters. Significant sensitivity to the various parameter perturbation configurations was found for all cloud species outputs. Unlike for the observable (column integral) quantities, the sensitivity for each cloud mixing ratio variable had relatively low time variability, and was associated primarily with the prevalence or not of the specific hydrometeor type in each phase of the simulated storm. For all outputs, with the exception of snow, the biases associated with the different configurations had the same sign and similar spatio-temporal patterns but different amplitudes. Since the microphysical parameters have the most significant and direct effect on the precipitating hydrometeors (rain, snow, and graupel), we analyze these first, followed by the non-precipitating cloud species (cloud droplets and pristine ice crystals).

The bias for the precipitating hydrometeors exhibited the highest variability among the configurations (Figures 5–7). For the rain (Figure 5) the positive bias (more rain than control) during the transition phase from convective to stratiform (90–120 min) was high for the SP-LN and SP-N configurations (respectively Figures 5a and 5b), moderate for the SP-NCov (Figure 5c) and negligible for the SP-NCovV and non-SP (respectively Figures 5d and 5e). This positive bias in SP-LN and SP-N configurations is consistent with the high bias in precipitation rate and liquid water path for these two experiments during this time period (respectively Figures 3b and 3d). The negative rain bias during the stratiform phase had the highest values associated with the SP-LN and lowest for the SP-NCovV, again consistent with the biases (or lack thereof) in PR and LWP in the last hour of the simulation (Figures 3b and 3d). The impact of all SP configurations on the graupel was mass reduction at heights between 6 and 8 km during the transition phase (Figure 6), where the control had the highest load (Figure 1b). The





Figure 6. As in Figure 5 for graupel.

loss of graupel was coupled with the rain increase for the SP-LN, SP-N and SP-NCov configurations (comparing Figures 6a-6c and 5a-5c) but not for the SP-NCovV (Figures 6 and 5d). The result indicates that the different microphysics processes were affected by the parameter perturbations between the different configurations. This was corroborated by the inspection of the differences between impacts of the configurations on the snow



Figure 7. As in Figure 5 for snow.





Figure 8. As in Figure 5 for pristine ice.

bias (Figure 7). As for the other hydrometeors the largest bias was associated with the SP-LN configuration (Figure 7a), however the smallest was exhibited by the SP-N configuration which also produced a different spatio-temporal pattern than the other configurations. While the bias for SP-LN, SP-Ncov, SP-NCovV and also non-SP experiments was characterized by a systematic vertical shift in the snow mass profile resulting in the dipole pattern (Figures 7a, 7c–7f), the SP-N experiment produced only a small systematic reduction of the snow mass at approximately the same height during the stratiform phase of the simulated storm (Figure 7b).

The variability of bias amplitudes for the non-precipitating ice and liquid between the SP configurations was smaller than for the precipitating hydrometeors. For the pristine ice the largest impacts were found during the convective phase where the associated microphysical processes were most active (Figure 8). For all configurations the bias was negative for the ice aloft, above about 8 km, and mostly positive below that. This is consistent with the high biases in precipitating hydrometeor contents below; less water vapor mass was available to be transported vertically and converted to pristine ice. Similarly, the impacts of the SP scheme on the cloud droplets for all configurations were significant only during the convective phase (Figure 9). The bias for cloud droplets was uniformly negative and of smaller relative amplitude than for other hydrometeors. Negative bias for cloud droplets was also consistent with the positive bias in precipitating hydrometeors. As for the precipitating hydrometeors the largest bias for the pristine ice and cloud droplets was found for the SP-LN configuration.

As would be expected the impacts of the SP configurations on the ensemble mean microphysics states had a close relationship to the ensemble biases in the observable states. For example, the negative peaks in the OSW bias (Figure 2f) during the convective and transition phases were directly related to the negative biases in the pristine ice and then graupel. Also, the mean loss of graupel mass was reflected in the high negative bias in the IWP (Figure 2d) and the positive in the OLW (Figure 2e). Similarly, the elevation of the snow mass (exhibited as the dipole pattern bias in Figure 7) caused the change of sign for the biases in the IWP and OLW (respectively Figures 2c and 2f) for the SP-LN and SP-NCov configurations which exhibited the significant bias amplitudes for that hydrometeor (Figures 8a and 8c). On the other hand the effect of the non-dipole and low amplitude snow bias for the SP-N configuration during the stratiform phase (Figure 8b) resulted in the low bias for the OSW (Figure 2f, red curves). Overall the results point to the complexity of understanding and ranking the configurations by the mean ensemble properties in the observable quantities alone. Similar to the results in the previous section they recommend using higher frequency evaluations (e.g., minutes to hourly) and also more diverse





Figure 9. As in Figure 5 for cloud droplets.

microphysics-sensitive observations (e.g., space and ground based radar observations) to enable a more accurate capture of different phases of the microphysics in order to better evaluate and understand the differences between the parameter uncertainty representations.

4. Summary and Conclusions

This study investigates sensitivity of the microphysics parameterization to configurations of the Stochastic Parameters (SP) schemes at the process level. For this purpose we use the generalized linear Gaussian autoregressive order-one stochastic model to represent the stochastic variability in the SP schemes. This formulation allows for exploration of the different configurations of the scheme, including: different parameter distributions, uncorrelated and correlated parameter perturbations, and different decorrelation times within the same algorithm. It is shown that the stochastic process models that are employed in the SPP (Ollinaho et al., 2017) and RP (McCabe et al., 2016) schemes commonly used in the global and regional ensemble systems in prior studies could be represented as special cases of the formulation used in this study. In the current study the stochastic model constitutes the SP scheme.

We apply the scheme to multiple microphysical parameters within the Lagrangian single column version (Posselt & Vukicevic, 2010) of the NASA Goddard Cumulus Ensemble (GCE) model (Tao et al., 2014) for an idealized representation of mid-latitude squall-line convection. This modeling framework was selected because it enables an investigation of the effect of changes in the microphysics parameters on the model output in isolation from any feedback to the cloud-scale dynamics. Also we were able to make use of the prior study results where the same model and test case were used to quantify multi-parameter uncertainty by means of multivariate MCMC inversions using simulated satellite observations of PR, LWP, IWP, OSW and OLW (Posselt & Vukicevic, 2010). This enabled using the estimates of multi-parameter distributions from the prior inversions as the basis for setting the statistical properties of parameter distributions used in the SP scheme. Additionally, the impacts of the non-stochastic and stochastic multi-parameter perturbations on the microphysics model solution and the related satellite-based observable quantities could be directly compared.

The sensitivity to the SP configurations using the 1D column model was carried out through ensemble sensitivity experiments described in Section 2.2 and summarized in Table 2. The experiments explored impacts of using



Log-Normal and Normal distributions, uncorrelated and correlated parameter perturbations, number of perturbed parameters, and different decorrelation time scales. The comparison of correlated versus uncorrelated perturbations was performed using only the Normal distribution. The experiments using the correlated perturbations included time-invariant and time-varying covariance estimates. The constant in time covariance was computed using the prior study inversion results, which considered the entire storm evolution. The time-varying covariance was estimated in this study by performing new MCMC inversions separately for the convective, transitional and stratiform phases of the simulated storm. For all experiments the theoretical mean value for each parameter was set to the constant parameter value used in the control version of the deterministic microphysics parameterization, the version that was used to simulate the observations for the inversions with the satellite observations. All configurations were applied to sets of 7 and 8 microphysical parameters and decorrelation times of 1, 3, 6 and 12 hr (Table 2). For all SP configurations the experiments involved 100-member ensemble simulations where each member was evolved with a different stochastic sequence of parameter perturbations. Additionally the same size ensemble simulations with randomly sampled constant-in-time parameters based on the empirical multi-parameter distribution from the prior study inversion were performed.

The impacts of different configurations of the SP scheme were measured with respect to the control simulation in terms of the ensemble RMSE, bias, and spread in the observation space, and in terms of ensemble mean bias for the the time evolving vertical profiles of rain, graupel, snow, pristine ice and cloud droplets.

The main findings are as follows:

- The ensemble diagnostics in the observation space exhibited high storm-phase-dependent time variability for all SP configurations and strong sensitivity to the choice of configuration.
- For all SP configurations, the temporal variability was most pronounced for the PR, OSW and OLW observation variables. The peaks in the bias (with respect to the control simulation) and spread for these observables were prominent during different phases of the storm depending on what microphysics processes were most impactful on the particular observable. For example, the bias and spread for OLW exhibited strong increases during the convective phase and then declined through the transition phase. For the IWP, LWP and AP the bias showed pronounced peaks during the convective-to-stratiform transition and stratiform phases, influenced by the changes in the ice microphysics. The ensemble spread for these observables, however, was characterized by low temporal variability and growth throughout the integration period.
- Concerning the sensitivity of the ensemble performance to the choice of SP configuration, the most influential factors were found to be the choice of the parameter distribution (normal vs. lognormal) and whether or not the parameter perturbations were correlated. Choice of temporal decorrelation time had little effect on the results of our experiments. For all observables the biases were overall largest for the configurations that used the univariate Log-Normal and Normal distributions (SP-LN and SP-N experiments) and smallest for the configurations with the storm-phase-dependent covariances (the SP-NCovV experiments). For the ensemble spread, the degree of sensitivity to configurations exhibited more dependence on the observable. The SP-NCov and SP-NCovV configurations exhibited the highest spread for the AP and IWP, whereas the SP-N and SP-LN configurations produced compatible or slightly higher spreads for the other observables.
- In terms of the RMSE-to-spread performance measure, the SP-NCovV experiments produced the most favorable ensemble response due to least bias and considerable spread. By the same measure the SP-LN configurations provided the least favorable ensembles.
- The experiments using constant in time parameters (the non-SP configurations) exhibited very different behavior from the SP experiments: for all observables the ensemble spread had slow growth throughout the simulations and was significantly smaller than for the SP experiments for the first 2 hr. The ensemble bias was small by design because the parameter samples were drawn from a joint empirical parameter distribution constrained by the simulated observations over the entire storm evolution and centered on the parameter mean values. The results suggest that time-invariant parameter perturbations do not support the variability of model responses due to different microphysics processes during the storm evolution.
- Significant sensitivity to the configurations was found for all microphysics state outputs. Unlike for the column integral observable quantities, the sensitivity for each cloud microphysical mass mixing ratio variable had relatively low time variability, and was associated primarily with the prevalence or not of the specific hydrometeor type in each phase of the simulated storm. For all outputs with the exception of the snow, the

biases (with respect to the control simulation) associated with the different configurations had the same sign and similar spatio-temporal patterns but different amplitudes.

- Overall, the bias in the microphysical states was highest for the SP-LN and SP-N configurations and lowest for SP-NCov and SP-NCovV. Most significant biases were exhibited for the rain, graupel and snow.
- The sensitivity to changing the number of parameters from 7 to 8 by adding the slope intercept for the rain particle size distribution was negligible overall except it caused higher time variability of the ensemble spread and RMSE for the precipitation rate.

These findings lead to the following main conclusions and recommendations.

- The impact of the stochastic parameter perturbations on the microphysics model ensemble solution is highly storm-phase dependent
- The correlated parameter perturbations informed by the process-level covariance estimates may reduce biased response by the microphysics parameterization and introduce more variability than the uncorrelated sampling using the univariate distributions
- Using the Log-Normal distribution or other skewed sampling may lead to undesirable biased response by the microphysics parameterization
- Using the appropriate mean and covariance estimates for the stochastic parameter sampling for the different phases of the microphysics dependent on the storm morphology is highly desirable
- This calls for the parameter estimation capabilities using the microphysics-sensitive observations preferably with hourly or less temporal frequency such as geostationary satellite and ground-based radar remote sensing. Such capability could be enabled by convective scale data assimilation systems

The primary caveat to this study, and the factor limiting its applicability in real-world prediction scenarios, is the use of a single column model framework. While this has enabled us to conduct a wide range of experiments and has enabled a more straightforward interpretation of the results, there is no feedback from parameter perturbations to the three-dimensional dynamics and thermodynamic state. This is also likely the reason there was little to no sensitivity of our results to changes in the decorrelation time scale; the 1D model evolution is controlled by the imposed water vapor source and vertical velocity profile. In previous parameter estimation studies, we have shown that the parameters that are most influential, the time during which they have an effect, and the inter-parameter covariances are quite similar between 1D and 3D models (compare results reported by Posselt and Vukicevic (2010) with Posselt (2016)). However, these studies considered only non-stochastic perturbation methods. The natural next step for our study is to employ the various SP methodologies in a three dimensional simulation framework and perhaps also to examine the ensemble spread due to initial conditions versus parameter perturbations as was done for static parameter perturbations by Posselt et al. (2019). We expect that the temporal autocorrelation length scale will have a much more substantial effect in a model that allows feedback from the parameterization to the thermodynamic and dynamic environment.

In addition, we have examined microphysical parameter perturbations in one particular type of weather system (deep convection). It would be useful to explore the influence of choices in SP scheme for other weather systems (e.g., extratropical cyclones and/or orographic precipitation). We also are aware that cloud microphysical schemes are not the only parameterizations with uncertainty. Other parts of the model (e.g., the planetary boundary layer parameterization; Aksoy et al., 2006) contain uncertain parameters and also would benefit from study of various SP scheme configurations. We plan to pursue these and other studies in the near future.

Appendix A: Relationship Between Generalized Linear Gaussian AR1 Formulation and the SPP and PR Schemes

In this appendix the correspondence between the SPP and RP schemes and the generalized linear Gaussian AR1 formulation used in this study is presented. By the generalized formulation the stochastic parameter perturbations are evolved according to the expression (Equation 1), repeated here for convenience

$$\Phi(t + \Delta t) - \underline{\Phi} = \alpha \left[\Phi(t) - \underline{\Phi} \right] + C_{\Phi}^{1/2} \varepsilon(t)$$
(A1)



where Φ is a vector stochastic quantity with mean $\underline{\Phi}$, α is the first order regression coefficient determined by the model time step dt and decorrelation time scale Λ , $C_{\Phi}^{1/2}\epsilon(t)$ is the innovation term where $C_{\Phi}^{1/2}$ is square root covariance matrix, and $\epsilon(t)$ is a N(0, I) random vector, with I being the identity matrix. Assuming no spatial variability for the parameter perturbations the SPP scheme formulation would derive directly from the expression (Equation A1) by using a diagonal variance matrix C_{Φ} and $\alpha = e^{-\frac{dt}{\Lambda}}$. As discussed in Section 2.2, the definition of the stochastic quantity would depend on the choice of statistical distribution for the parameter variability within a range of permissible values. For the scheme application in the ECMWF medium-range ensemble forecasting system Ollinaho et al. (2017) used the Long Normal distribution. For this distribution as is shown in Section 2.2. the vector Φ consists of the quantities $\varphi = Ln(\frac{p'}{p})$, where p and p' are, respectively, the unperturbed and perturbed parameter values for each parameter to be perturbed. Ollinaho et al. (2017) considered two options for φ based on the following two criteria: $\underline{p'} = p$ and $p'_{median} = p$. The first criterion corresponds to the SP-LN configuration in this study with $\underline{\varphi} = -\frac{\sigma_{\varphi}^2}{2}$ and $\sigma_{\varphi}^2 = Ln\left(\frac{\sigma_p^2}{p^2} + 1\right)$, where σ_p^2 is predefined parameter variance, whereas the second criterion corresponds to $\varphi = 0$.

The recent applications of the SPP scheme with the regional models (Jankov et al., 2019; Stanford et al., 2019; Thompson et al., 2021; Wang et al., 2019) considered the Normal distribution for the SPP using the multiplicative stochastic noise formulation for the relationship between the perturbed and unperturbed parameter values according to $p' = (1 + \varphi)p$. For this formulation the expression (Equation A1) would apply to $\varphi = \frac{p'}{p} - 1$, $\underline{\varphi} = 0$ and $\sigma_{\varphi} = \frac{\sigma_p}{p}$, equivalent to the SP-N configuration used in this study. When the spatial variability of parameter perturbations is considered and represented by the Gaussian correlated noise pattern in 2D horizontal space of the prediction model, as was done for all prior applications of the SPP scheme, the stochastic quantity φ would involve spectral coefficients of the spatial noise pattern.

The PR scheme (Hermoso et al., 2021; McCabe et al., 2016) assumes no spatial variability of the parameters and makes use of the uniform distribution for the stochastic variability according to

$$P(t+1) = \alpha P(t) + 2\eta \sqrt{1 - \alpha^2}$$

where $\alpha = 1 - \frac{dt}{\Lambda}$ and η is a random quantity with uniform distribution in the interval [-1,1], could be represented by the model (Equation A1) using $\varphi = \frac{p'}{p} - 1$ with mean $\underline{\varphi} = 0$. To represent the sampling equivalent to using the bounded uniform distribution the variance may be set to $\sigma_{\varphi} = 2\sigma^* \sqrt{1 - \alpha^2}$, where σ^* would be a large value applicable to all parameters. The RP scheme and this configuration are not exact equivalents due to approximating the uniform distribution by the Normal with a large variance, but the correspondence would be close for all practical purposes assuming application of the appropriate boundary conditions (the range of parameter values). The RP configuration was not explicitly considered in the sensitivity experiments in this study because it is similar to the SP-N configuration (Table 2).

Data Availability Statement

The data used in creation of this manuscript include: simulations by the 1D column microphysics model used in this study, diagnostics derived from the model simulations and results from Bayesian inversions performed in Posselt and Vukicevic (2010) and in this study using the same model and simulated column integral satellite observations. The data are available at https://doi.org/10.5281/zenodo.5736940. A guide to content and format of the archived data is provided in the Supporting information. The graphics were generated using Matplotlib (https://matplotlib.org) and Seaborn (https://seaborn.pydata.org) software libraries. The 1D column microphysics model used in this study is a single column version of the NASA Goddard Cumulus Ensemble (GCE) model (Tao et al., 2014), described in Posselt and Vukicevic (2010). The Bayesian inversion model used in this study is presented in Posselt and Vukicevic (2010).



Acknowledgments

This study was supported by funding from the European Union's Horizon 2020 research and innovation programme under the Marie Sklodowska-Curie grant agreement No 101003352 at the Faculty of Physics, University of Belgrade, Serbia. This research was partially carried out through A. Stankovic's master thesis research while at the Faculty of Physics, University of Belgrade, Serbia.

References

- Aksoy, A., Zhang, F., & Nielsen-Gammon, J. W. (2006). Ensemble-based simultaneous state and parameter estimation in a two-dimensional sea-breeze model. *Monthly Weather Review*, 134(10), 2951–2970. https://doi.org/10.1175/mwr3224.1
- Baker, L., Rudd, A., Migliorini, S., & Bannister, R. (2014). Representation of model error in a convective-scale ensemble prediction system. *Nonlinear Processes in Geophysics*, 21(1), 19–39. https://doi.org/10.5194/npg-21-19-2014
- Bengtsson, L., Bao, J., Pegion, P., Penland, C., Michelson, S., & Whitaker, J. (2019). A model framework for stochastic representation of uncertainties associated with physical processes in NOAA's next generation global prediction system (NGGPS). *Monthly Weather Review*, 147(3), 893–911. https://doi.org/10.1175/MWR-D-18-0238.1
- Bengtsson, L., Körnich, H., Källén, E., & Svensson, G. (2011). Large-scale dynamical response to subgrid-scale organization provided by cellular automata. Journal of the Atmospheric Sciences, 68(12), 3132–3144. https://doi.org/10.1175/JAS-D-10-05028.1
- Berner, J., Achatz, U., Batté, L., Bengtsson, L., Cámara, A. d., Christensen, H. M., et al. (2017). Stochastic parameterization: Toward a new view of weather and climate models. *Bulletin America Meteorology Social*, 98(3), 565–588. https://doi.org/10.1175/BAMS-D-15-00268.1
- Berner, J., Fossell, K. R., Ha, S.-Y., Hacker, J. P., & Snyder, C. (2015). Increasing the skill of probabilistic forecasts: Understanding performance improvements from model-error representations. *Monthly Weather Review*, 143(4), 1295–1320. https://doi.org/10.1175/MWR-D-14-00091.1 Berner, J., Ha, S.-Y., Hacker, J. P., Fournier, A., & Snyder, C. (2011). Model uncertainty in a mesoscale ensemble prediction system: Stochastic
- versus multiphysics representations. *Monthly Weather Review*, 139(6), 1972–1995. https://doi.org/10.1175/2010MWR3595.1
- Bouttier, F., Vié, B., Nuissier, O., & Raynaud, L. (2012). Impact of stochastic physics in a convection-permitting ensemble. *Monthly Weather Review*, 140(11), 3706–3721. https://doi.org/10.1175/MWR-D-12-00031.1
- Buizza, R., Miller, M., & Palmer, T. N. (1999). Stochastic representation of model uncertainties in the ECMWF ensemble prediction system. *Quarterly Journal of the Royal Meteorological Society*, 125(560), 2887–2908. https://doi.org/10.1002/qj.49712556006
- Charron, M., Pellerin, G., Spacek, L., Houtekamer, P. L., Gagnon, N., Mitchell, H. L., & Michelin, L. (2010). Toward random sampling of model error in the Canadian ensemble prediction system. *Monthly Weather Review*, 138(5), 1877–1901. https://doi.org/10.1175/2009mwr3187.1
- Christensen, H. M., Berner, J., Coleman, D., & Palmer, T. N. (2017). Stochastic parameterization and the El Niño–south-ern oscillation. Journal of Climate, 30(1), 17–38. https://doi.org/10.1175/JCLI-D-16-0122.1
- Christensen, H. M., Moroz, I. M., & Palmer, T. N. (2015). Stochastic and perturbed parameter representations of model uncertainty in convection parameterization. Journal of the Atmospheric Sciences, 72(6), 2525–2544. https://doi.org/10.1175/JAS-D-14-0250.1
- Craig, G. C., & Cohen, B. G. (2006). Fluctuations in an equilibrium convective ensemble. Part I: Theoretical formulation. Journal of the Atmospheric Sciences, 63(8), 1996–2004. https://doi.org/10.1175/jas3709.1
- Davini, P., von Hardenberg, J., Corti, S., Christensen, H. M., Juricke, S., Subramanian, A., et al. (2017). Climate SPHINX: Evaluating the impact of resolution and stochastic physics parameterisations in climate simulations. *Geoscientific Model Development Discussions*, 10(3), 1383–1402. https://doi.org/10.5194/gmd-10-1383-2017
- Debussche, A., Glatt-Holtz, N., Temam, R., & Ziane, M. (2012). Global existence and regularity for the 3D stochastic primitive equations of the ocean and atmosphere with multiplicative white noise. *Nonlinearity*, 25(7), 2093–2118. https://doi.org/10.1088/0951-7715/25/7/2093
- Grunwald, G. K. R., Hyndman, J., Tedesco, L., & Tweedie, R. L. (2000). Non-Gaussian conditional linear AR (1) models. Australian & New Zealand Journal of Statistics, 42(4), 479–495. https://doi.org/10.1111/1467-842X.00143
- Hermoso, A., Homar, V., & Plant, R. S. (2021). Potential of stochastic methods for improving convection-permitting ensemble forecasts of extreme events over the Western Mediterranean. Atmospheric Research, 257, 105571. https://doi.org/10.1016/j.atmosres.2021.105571
- Jankov, I., Beck, J., Wolff, J., Harrold, M., Olson, J., Smirnova, T., et al. (2019). Stochastically perturbed parameterizations in an HRRR-based ensemble. *Monthly Weather Review*, 147(1), 153–173. https://doi.org/10.1175/MWR-D-18-0092.1
- Jankov, I., Berner, J., Beck, J., Jiang, H., Olson, J. B., Grell, G., et al. (2017). A performance comparison between multiphysics and stochastic approaches within a North American RAP ensemble. *Monthly Weather Review*, 145(4), 1161–1179. https://doi.org/10.1175/MWR-D-16-0160.1
- Khain, A. P., Beheng, K. D., Heymsfield, A., Korolev, A., Krichak, S. O., & Levin, Z. (2015). Representation of microphysical processes in cloud-resolving models: Spectral (bin) microphysics versus bulk parameterization. *Review of Geophysics*, 53(2), 247–322. https://doi. org/10.1002/2014rg000468
- Kober, K., & Craig, G. C. (2016). Physically based stochastic perturbations (PSP) in the boundary layer to represent uncertainty in convective initiation. Journal of the Atmospheric Sciences, 73(7), 2893–2911. https://doi.org/10.1175/JAS-D-15-0144.1
- Lang, S., Tao, W.-K., Cifelli, R., Olson, W., Halverson, J., Rutledge, S., & Simpson, J. (2007). Improving simulations of convective systems from TRMM LBA: Easterly and westerly regimes. *Journal of the Atmospheric Sciences*, 64(4), 1141–1164. https://doi.org/10.1175/JAS3879.1
- Leutbecher, M., Lock, S., Ollinaho, P., Lang, S., Balsamo, G., Bechtold, P., et al. (2017). Stochastic representations of model uncertainties at ECMWF: State of the art and future vision. *Quarterly Journal of the Royal Meteorological Society*, 143(707), 2315–2339. https://doi.org/10.1002/qj.3094
- Lin, Y.-L., Farley, R. D., & Orville, H. D. (1983). Bulk parameterization of the snow field in a cloud model. Journal of Climate and Applied Meteorology, 22(6), 1065–1092. https://doi.org/10.1175/1520-0450(1983)022<1065:bpotsf>2.0.co;2
- McCabe, A., Swinbank, R., Tennant, W., & Lock, A. (2016). Representing model uncertainty in the Met Office convection-permitting ensemble prediction system and its impact on fog forecasting. *Quarterly Journal of the Royal Meteorological Society*, 142(700), 2897–2910. https://doi. org/10.1002/qj.2876
- Ollinaho, P., Lock, S., Leutbecher, M., Bechtold, P., Beljaars, A., Bozzo, A., et al. (2017). Towards process-level representation of model uncertainties: Stochastically perturbed parametrizations in the ECMWF ensemble. *Quarterly Journal of the Royal Meteorological Society*, 143(702), 408–422. https://doi.org/10.1002/qj.2931
- Palmer, T. N. (2001). A nonlinear dynamical perspective on model error: A proposal for non-local stochastic-dynamic parameterization in weather and climate prediction. *Quarterly Journal of the Royal Meteorological Society*, 127(572), 279–304. https://doi.org/10.1002/qj.49712757202
- Palmer, T. N., Buizza, R., Doblas-Reyes, F., Jung, T., Leutbecher, M., Shutts, G., et al. (2009). Stochastic parametrization and model uncertainty. *ECMWF Tech. Memo.*, 598, 44. Retrieved from www.ecmwf.int/sites/default/files%20/elibrary/2009/11577-stochastic-parametrization%20 -and-model-uncertainty.pdf
- Palmer, T. N., Shutts, G. J., Hagedorn, R., Doblas-Reyes, F. J., Jung, T., & Leutbecher, M. (2005). Representing model uncertainty in weather and climate prediction. Annual Review of Earth and Planetary Sciences, 33(1), 163–193. https://doi.org/10.1146/annurev.earth.33.092203.122552
- Posselt, D., & Vukicevic, T. (2010). Robust characterization of model physics uncertainty for simulations of deep moist convection. *Monthly Weather Review*, 138(5), 1513–1535. https://doi.org/10.1175/2009MWR3094.1
- Posselt, D., Vukicevic, T., & Stankovic, A. (2021). Data used in the publication: Sensitivity of modeled microphysics to stochastically perturbed parameters [Dataset]. Zenodo, https://doi.org/10.5281/zenodo.5736940



- Posselt, D. J. (2016). A bayesian examination of deep convective squall-line sensitivity to changes in cloud microphysical parameters. *Journal of the Atmospheric Sciences*, 73(2), 637–665. https://doi.org/10.1175/JAS-D-15-0159.1
- Posselt, D. J., & Bishop, C. H. (2012). Nonlinear parameter estimation: Comparison of an ensemble kalman smoother with a Markov chain Monte Carlo algorithm. *Monthly Weather Review*, 140(6), 1957–1974. https://doi.org/10.1175/MWR-D-11-00242.1
- Posselt, D. J., He, F., Bukowski, J., & Reid, J. S. (2019). On the relative sensitivity of a tropical deep convective storm to changes in environment and cloud microphysical parameters. *Journal of the Atmospheric Sciences*, 76(4), 1163–1185. https://doi.org/10.1175/JAS-D-18-0181.1
- Posselt, D. J., Hodyss, D., & Bishop, C. H. (2014). Errors in ensemble kalman smoother estimates of cloud microphysical parameters. *Monthly Weather Review*, 142(4), 1631–1654. https://doi.org/10.1175/MWR-D-13-00290.1
- Romine, G. S., Schwartz, C. S., Berner, J., Fossell, K. R., Snyder, C., Anderson, J. L., & Weisman, M. L. (2014). Representing fore-cast error in a convection-permitting ensemble system. *Monthly Weather Review*, 142(12), 4519–4541. https://doi.org/10.1175/MWR-D-14-00100.1
- Rutledge, S. A., & Hobbs, P. V. (1983). The mesoscale and microscale structure and organization of clouds and precipitation in midlatitude cyclones. VIII: A model for the "seeder-feeder" process in warm-frontal rainbands. *Journal of the Atmospheric Sciences*, 40(5), 1185–1206. https://doi.org/10.1175/1520-0469(1983)040<1185:tmamsa>2.0.CO;2
- Rutledge, S. A., & Hobbs, P. V. (1984). The mesoscale and microscale structure and organization of clouds and precipitation in midlatitude cyclones. Part XII: A diagnostic modeling study of precipitation development in narrow cold frontal rainbands. *Journal of the Atmospheric Sciences*, 41(20), 2949–2972. https://doi.org/10.1175/1520-0469(1984)041<2949:tmamsa>2.0.CO;2
- Sanchez, C., Williams, K. D., & Collins, M. (2015). Improved sto-chastic physics schemes for global weather and climate models. *Quart. J. Roy. Meteor. Soc.*, 142(694), 147–159. https://doi.org/10.1002/qj.2640
- Shutts, G., & Pallares, A. C. (2014). Assessing parametrization uncertainty associated with 1036 horizontal resolution in numerical weather prediction models. *Philosophical Transactions of the Royal Society A: Mathematical, Physical and Engineering Sciences*, 372(2018), 20130284. https://doi.org/10.1098/rsta.2013.0284
- Stanford, M. W., Morrison, H., Varble, A., Berner, J., Wu, W., McFarquhar, G., & Milbrandt, J. (2019). Sensitivity of simulated deep convection to a stochastic ice microphysics framework. *Journal of Advances in Modeling Earth Systems*, 11, 3362–3389. https://doi. org/10.1029/2019MS001730
- Tao, W.-K., Chern, J., Iguchi, T., Lang, S., Lee, M.-J., Li, X., et al. (2019). Microphysics in goddard multi-scale modeling systems: A review. In Current trend in the representation of physical processes in weather and climate models (pp. 253–316). Springer Nature. https://doi. org/10.1007/978-981-13-3396-5_14
- Tao, W. K., Lang, S., Zeng, X., Li, X., Matsui, T., Mohr, K., et al. (2014). The goddard Cumulus ensemble (GCE) model: Improvements and applications for studying precipitation processes. *Atmospheric Research*, 143, 392–424. https://doi.org/10.1016/j.atmosres.2014.03.005
- Thompson, G., Berner, J., Frediani, M., Otkin, J. A., & Griffin, S. M. (2021). A stochastic parameter perturbation method to represent uncertainty in a microphysics scheme. *Monthly Weather Review*, *149*(5), 1481–1497. https://doi.org/10.1175/MWR-D-20-0077.1
- van Lier-Walqui, M., Vukicevic, T., & Posselt, D. J. (2012). Quantification of cloud microphysical parameterization uncertainty using radar reflectivity. *Monthly Weather Review*, 140(11), 3442–3466. https://doi.org/10.1175/MWR-D-11-00216.1
- van Lier-Walqui, M., Vukicevic, T., & Posselt, D. J. (2014). Linearization of microphysical parameterization uncertainty using multiplicative process perturbation parameters. *Monthly Weather Review*, 142(1), 401–413. https://doi.org/10.1175/MWR-D-13-00076.1
- Wang, S., Qiao, X., Min, J., & Zhuang, X. (2019). The impact of stochastically perturbed parameterizations on tornadic supercell cases in east China. Monthly Weather Review, 147(1), 199–220. https://doi.org/10.1175/MWR-D-18-0182.1
- Weisheimer, A., Corti, S., Palmer, T. N., & Vitart, F. (2014). Addressing model error through atmospheric stochastic physical parametrizations: Impact on the coupled ECMWF seasonal forecasting system. *Philosophical Transactions of the Royal Society A: Mathematical, Physical and Engineering Sciences*, A372(2018). 2013029. https://doi.org/10.1098/rsta.2013.0290
- Yussouf, N., & Stensrud, D. J. (2012). Comparison of single-parameter and multiparameter ensembles for assimilation of radar observations using the ensemble Kalman filter. *Monthly Weather Review*, 140(2), 562–586. https://doi.org/10.1175/MWR-D-10-05074.1